Author's response to reviews

Title: I Move: systematic development of a web-based computer tailored physical activity intervention, based on motivational interviewing and self-determination theory.

Authors:

Stijn AH Friederichs (stijn.friederichs@ou.nl)
Anke Oenema (a.oenema@maastrichtuniversity.nl)
Catherine Bolman (catherine.bolman@ou.nl)
Janneke Guyaux (janneke.guyaux@ou.nl)
Hilde M van Keulen (hilde.vankeulen@tno.nl)
Lilian Lechner (lilian.lechner@ou.nl)

Version: 5
Date: 20 February 2014

Author's response to reviews:

Dear editorial board of BMC Public Health,

You requested additional information, which I will address below:

1. Ethical and Funding Approval Documentation

I sent copies of all ethical approval and funding approval for your records to BMCSeriesEditorial@biomedcentral.com. I also included translated versions of the relevant parts. All documents and as well as the document with the translations are signed by a contactable person at our institution.

2. Funding

Our study protocol has undergone peer review by the Netherlands Organisation for Health Research and Development (ZonMW) before being awarded with a grant.

3. Study status

The study described in the protocol is ongoing, as we are still collecting data.

4. Related Articles

No publications of this study have been submitted or published, since we are still collecting the data.

Yours sincerely, on behalf of all co-authors,